Author's response to reviews

Title: The incidence of co-morbidities related to obesity and overweight: A systematic review and meta-analysis

Authors:

Daphne P Guh (daphne@sm.hivnet.ubc.ca)
Wei Zhang (wzhang@cheos.ubc.ca)
Nick Bansback (nbansback@cheos.ubc.ca)
Zubin Amarsi (zamarsi@arthritisresearch.ca)
C. Laird Birmingham (clbirm@interchange.ubc.ca)
Aslam H Anis (aslam.anis@ubc.ca)

Version: 6 Date: 16 March 2009

Author's response to reviews:

Two small typographical changes have been made.